Author’s response to reviews

Title: Abnormal glucose tolerance and insulin resistance in polycystic ovary syndrome amongst the Taiwanese population- not correlated with insulin receptor substrate-1 Gly972Arg/Ala513Pro polymorphism

Authors:

Ta-Chin Lin (tachin@ksts.seed.net.tw)
Jui-Mei Yen (yenjoyce24@hotmail.com)
Kum-Bing Gong (quinn2@sinamail.com)
Tsung-Cheng Kuo (tckuo@kgh.com.tw)
Dong-Chi Ku (imwu@mail.chna.edu.tw)
Shu-Fen Liang (imwu@mail.chna.edu.tw)
Ming-Jiuan Wu (imwu@mail.chna.edu.tw)

Version: 5 Date: 3 April 2006

Author’s response to reviews:

Dear Editorials:
We had corrected some mistakes as below.
Page 8 line 16: dioxynucleoside biphosphates -> deoxynucleoside triphosphates
Page 10 line 14: 21.1+/-8.3 -> (21.1+/-8.3)
Page 10 line 15: 24.4+/-17.2 -> (24.4+/-17.2)
Page 14 line 3: EL Mkadem -> El Mkadem
Page 16 line 1: sixty subjects -> ninety-two subjects
Page 16 line 18: pf NIDDM -> of NIDDM